Author’s response to reviews

Title: Atrophic inferior vena cava is a marker of chronicity of intra-filter and inferior vena cava thrombosis: based on CT findings

Authors:
Liang Chen (Huly_nj@163.com)
Wanyin Shi (shwy110@163.com)
Jianping Gu (gjp_nj@163.com)
Xu He (hexu_nj@163.com)
Wensheng Lou (louwensheng_nj@163.com)

Version: 4 Date: 18 Mar 2018

Author’s response to reviews:

Cover Letter

To the Editors and Reviewers:

Thank you very much for accepting the corrections we've made. Sorry for one mistake left in the text. We have revised it in accordance with your comment. We hope the revised manuscript meets the acceptance standards of the BMC Cardiovascular Disorders. Please see below the responses to your comments and the revisions made.

-Dalibor Musil (Reviewer 2): I accept all corrections. The work well illustrates the development from acute IVC thrombosis to chronic IVC occlusion in patients with lower limb DVT and IVC filter placement, although the some limitations are retrospective study and small sample of patients. Only the comment: Seven filters were placed in the IVC via the right femoral vein because of a left-sided DVT. Only one filter was placed via the right side because of a left-sided DVT..... error marking the right and left side (p. 7)
-Reply: Thank you again for accepting the corrections we've made. And sorry for the mistake left in the text, which we have revised as below:

Seven filters were placed in the IVC via the right femoral vein because of a left-sided DVT. Only one filter was placed via the left side because of a right-sided DVT.

To the Reviewers: Thank you again for your reviews and helpful suggestions. We hope that our responses have addressed your concerns. Please let us know if you require further clarification.